Author's response to reviews

Title: Pre-dialysis patients' perceived autonomy, self-esteem and labor participation: Associations with illness perceptions and treatment perceptions. A cross-sectional study

Authors:

Daphne L Jansen (d.jansen@nivel.nl)
Diana C Grootendorst (d.c.grootendorst@lumc.nl)
Mieke Rijken (m.rijken@nivel.nl)
Monique JWM Heijmans (m.heymans@nivel.nl)
Ad A Kaptein (a.a.kaptein@lumc.nl)
Elisabeth W Boeschoten (els@hansmakinstituut.nl)
Friedo W Dekker (f.w.dekker@lumc.nl)
PREPARE-2 Study Group (d.jansen@nivel.nl)

Version: 6 Date: 3 December 2010

Author's response to reviews: see over
Dear Editor:

Re; “Pre-dialysis patients’ perceived autonomy, self-esteem and labor participation: Associations with illness perceptions and treatment perceptions. A cross-sectional study”, 
*manuscript id*: 1336861601393280.

We have checked the manuscript and have made some small changes (by means of track changes). Hereby we send the revised version of the manuscript.

Sincerely,
also on behalf of the co-authors,
Daphne Jansen